Author's response to reviews

Title: "Done more for me in a fortnight than anybody done in all me life." How welfare rights advice can help people with cancer

Authors:

Suzanne Moffatt (s.m.moffatt@ncl.ac.uk)
Emma Noble (e.j.noble@ncl.ac.uk)
Catherine Exley (catherine.exley@ncl.ac.uk)

Version: 8 Date: 2 September 2010

Author's response to reviews:

The final corrected version is now uploaded, filename 'Moffattetal2ndsept2010FINALCORRECTEDPROOFS.doc'.
The title and abstract are now the same on the paper and the electronic submission.
Thank you for your help with the paper.